Author’s response to reviews

Title: Predictive value of gamma-glutamyl transpeptidase to lymphocyte count ratio in hepatocellular carcinoma patients with microvascular invasion

Authors:

Hongxing Zhang (563707066@qq.com)
Yu Zhou (961001841@qq.com)
Yicheng Li (673336039@qq.com)
Wanying Qin (493900281@qq.com)
Yunhua Zi (1983698302@qq.com)
Yulan Liu (1152482457@qq.com)
Xiaoying Qiu (1402137663@qq.com)
Hongyuan Xu (2107025705@qq.com)
Weijia Liao (liaoweijia288@163.com)
Zhaoquan Huang (huang788766@163.com)

Version: 2 Date: 08 Feb 2020

Author’s response to reviews:

Dear editor:

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate editor and reviewers very much for their positive and constructive comments and suggestions on our manuscript (BCAN-D-19-03722). My responses to the comments are described as follows. We hope that our revised manuscript is now suitable for your requirements.

Sincerely
Weijia Liao.
E-mail: liaoweijia288@163.com

Review report
Manuscript ID number:
BCAN-D-19-03722
Title of paper:
Predictive value of gamma-glutamyl transpeptidase to lymphocyte count ratio in hepatocellular carcinoma patients with microvascular invasion

Here are responses to the comments:

Comment #1:
Currently, the contributions of authors [WQ, YZ, YL, XQ, and HX] do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions.
Response: We have corrected this part. Thank you for pointing out.

Comment #2
To ensure that both the abstract in your main text and in the submission system match, please include a keywords section at the end of the abstract in the submission system. Please ensure that you include a 'Declarations' heading before your declarations statements, after your list of abbreviations.
Response: All the suggestions of editorial formatting have been incorporated.

Comment #3
Please remove your [Editing certificate] and [Response to comments] supplementary files from the file inventory, as this is no longer required at this stage of the editorial process.
Response: Yes, we did.

Comment #4
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response: Yes, we did.

I would be happy to make any further changes that may be required.